**Author’s response to reviews**

**Title:** The immunomodulatory activities of licorice polysaccharides (Glycyrrhiza uralensis Fisch.) in CT 26 tumor-bearing Mice

**Authors:**

Peter Ayeka (amwogahpeter@yahoo.com)

Yuhong Bian (bianyuhong_2012@163.com)

Peter Githaiga (pmwitari67@yahoo.com)

Ying Zhao (zyzy_1@163.com)

**Version:** 3 **Date:** 10 Nov 2017

**Author’s response to reviews:**

1. We have included the email address of all authors on the title page

2. We have added in the methods section, a sub section titled "euthanasia of mice" line 155-159

3. We have also removed figure captions in the two figure files and put the captions at the end of the manuscript, line 557-570

4. We have added a Dot plot figure for flow cytometry data